Author's response to reviews

Title: Purchasing and Using Personal Emergency Response Systems (PERS): How Decisions are Made by Community-Dwelling Seniors in Canada

Authors:

Alexandra C McKenna (mckenna.alexandra@gmail.com)
Marita Kloseck (mkloseck@uwo.ca)
Richard G Crilly (richard.crilly@sjhc.london.on.ca)
Jan Polgar (jpolgar@uwo.ca)

Version: 2 Date: 19 February 2014

Author's response to reviews: see over
BMC Geriatrics  
Attention: Editorial Staff

Dear Sir or Madam:

RE: MS: 1920793990102888

We have revised the above manuscript and included the Trial registration: Current Controlled Trials ISRCTN44810824 statement at the end of the abstract as requested. We have also added the informed consent statement at the end of the Methods section.

We look forward to your response.

Yours truly

Richard Crilly, MD  
Associate Professor  
Faculty of Medicine, Division of Geriatric Medicine  
University of Western Ontario

RC: jmcc  
Encls.